Author's response to reviews

Title: Cholecystomucoclasis: Revaluation of Safety and Validity in the Aged Population

Authors:

Tomoya Tsukada (tkd_tmy@nifty.com)
Tatsuo Nakano (nakano@asanogawa-gh.or.jp)
Takashi Miyata (ryutami5383917@gmail.com)
Shozo Sasaki (syozossjp@yahoo.co.jp)
Tetsuo Ohta (ohtat@staff.kanazawa-u.ac.jp)

Version: 2 Date: 6 March 2012

Author's response to reviews: see over
March 6, 2012

Mr. Timothy Shipley  
Executive Editor  
BMC Gastroenterology

To the Editors.

The comments of the editors have been helpful in allowing us to revise our manuscript.

-Ethics:
We have made more detailed statement " This study was approved by the Institutional Review Board (IRB) of the Asanogawa General Hospital (AGH-IRB No. 22)." in Method part on page10.

-Acknowledgements:
We have made more detailed statement " There is no one who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data." in Acknowledgements part on page13-14.

Thank you for your consideration. I look forward to hearing from you.

Sincerely,

Tomoya Tsukada  
Department of Gastroenterological Surgery, Division of Cancer Medicine  
Graduate School of Medical Science, Kanazawa University  
13-1 Takara-machi, Kanazawa, Ishikawa 920-8641, Japan  
Phone: 81-76-265-2362  
Fax: 81-76-234-4260  
E-mail: tkd_tmy@nifty.com